Author's response to reviews

Title: The effect of long term combined yoga practice on the basal metabolic rate of healthy Adults

Authors:

M S Chaya (chayapu@hotmail.com)
Kurpad A V (a.kurpad@iphcr.res.in)
Nagendra H R (hrnagendra@rediffmail.com)
Nagarathna R (rn44@rediffmail.com)

Version: 7 Date: 28 August 2006

Author's response to reviews:

Dear Editor,
Thank you for the email requesting corrections. We are uploading the corrected version, as per the guidelines noted in your email as well in the instructions to author, (especially the fonts size and headings).

If need be we will happy to provide any other information.
Thank you very much for the support.
Chaya